Author's response to reviews

Title: Public preferences for engagement in Health Technology Assessment decision-making: protocol of a mixed methods study

Authors:

Sally wortley (sally.wortley@sydney.edu.au)
Allison Tong (allison.tong@sydney.edu.au)
Emily Lancsar (emily.lancsar@monash.edu)
Glenn Salkeld (Glenn.Salkeld@sydney.edu.au)
Kirsten Howard (kirsten.howard@unisa.ey.edu.au)

Version: 5 Date: 16 June 2015

Author's response to reviews:

16th June 2015

To the Editor,

Please find attached our manuscript MS: 201878552714591 ‘Public preferences for engagement in Health Technology Assessment decision-making: protocol of a mixed methods study’ that addresses the revisions requested in the email dated 12th June 2015.

I can confirm that we have made the following amendments
- abstract line 8: ‘pubic’ has now been changed to ‘public’
- methodology section has been renamed Methods/Design

A revised manuscript with the changes has been submitted.

We can confirm that this paper has not been published nor submitted for publication elsewhere. We can also confirm that that there are no competing/conflicts of interests and that each author contributed to the conception writing of the paper; each has approved the version being submitted.

Please address all correspondence to sally.wortley@sydney.edu.au

We look forward to hearing from you at your earliest convenience.

Regards

Sally Wortley